ORIGINAL ARTICLE



Influence of climate-smart technologies on the success of livestock donation programs for smallholder farmers in Rwanda

John M. Kandulu¹ · Alec Zuo¹ · Sarah Wheeler¹ · Theogene Dusingizimana² · Mizeck G. G. Chagunda³

Received: 17 March 2023 / Accepted: 28 February 2024 / Published online: 15 March 2024 © The Author(s) 2024

Abstract

Climate change threatens the livelihoods of Sub-Saharan African farmers through increased droughts. Livestock donation programs offer a potential solution, but their effectiveness under climate stress remains unclear. This study assesses the economic viability of integrating climate-smart technologies (cowsheds and biogas plants) into these programs in Rwanda. Using a stochastic benefit–cost analysis from the beneficiary perspective, we evaluate the net gains for households receiving heifers compared to the current program. Our findings reveal that integrating climate-smart technologies significantly enhances economic viability. Households with cows and climate-smart technologies can possibly realise net benefits 3.5 times higher than the current program, with benefit–cost ratios reaching 5:1. Beyond economic benefits, adopting biogas reduces deforestation, greenhouse gas emissions, and respiratory illness risks. This study demonstrates that integrating climate-smart technologies into livestock donation programs can generate positive economic, environmental, and health benefits, leading to more resilient and sustainable smallholder systems. However, overcoming implementation challenges requires tailored policy packages addressing local barriers.

Keywords Aid effectiveness \cdot Sustainable economic development \cdot Low- and mediumincome countries \cdot Poverty reduction

John M. Kandulu john.kandulu@adelaide.edu.au

¹ School of Economics and Public Policy, The University of Adelaide, 10 Pulteney St, Adelaide, South Australia 5005, Australia

² College of Agriculture, Animal Science and Veterinary Medicine, University of Rwanda, P.O. Box 210, Musanze, Kigali, Rwanda

³ Animal Breeding and Husbandry in the Tropics and Subtropics, University of Hohenheim, Garben Street, 17 Instituts- Und Hörsaalgebäude, -112, Stuttgart, Germany

1 Introduction

Sustainable development goals (UN General Assembly 2015) recognise the critical role of agricultural development in poverty reduction and food security for marginalised rural populations in Sub-Saharan Africa. Agricultural development is a key poverty reduction strategy due to its potential for inclusivity compared to non-agricultural sectors (Corral et al. 2017; FAO 2018; World Bank 2016; 2018).

Sub-Saharan Africa's rural communities are particularly vulnerable to the impacts of climate change, including frequent droughts, intensified floods and storms, and variable rainfall, which undermine food security and compromise the viability of smallholder croplivestock production systems (Abebe et al. 2022; 2023; Adesete et al. 2023; Baptista et al. 2022; Bedasa and Deksisa 2024). Sub-Saharan Africa's vulnerability to variable climate is heightened by its dependence on rain-fed, low-input, and small-scale cropping systems (Omotoso et al. 2023; Njoya et al. 2022). Increasing rates of greenhouse gas (GHG) emissions from agricultural production are projected to further exacerbate the frequency and severity of changes in weather patterns (The Intergovernmental Panel on Climate Change (IPCC), 2007, 2023). In response to the intensifying climate change threat faced by smallholder agricultural systems in low- and middle-income countries, there is a need to prioritise effective adaptation and mitigation efforts (Hansen et al. 2019; Madamombe et al. 2024; Shikuku et al. 2017) and address growth inequality issues (Manero et al. 2020). Improving access to climate-smart technological innovations can mitigate the impact of climate change on food production and the livelihoods of the region's population (Gashure et al. 2022).

Staple crop yields could decrease by up to 10–15% due to projected changes in temperature and precipitation by 2050 in Rwanda (Austin et al. 2020), highlighting the need to prioritise support for food security and climate resilience in Rwanda's agricultural sector (Mperejekumana et al. 2024; Niyitanga et al. 2015). As the impact of climate change poses a threat to Rwanda's agricultural economy, the adoption of climate-smart innovations, such as diversified livestock and crop systems and the use of renewable fuels, holds the potential to enhance the resilience of food production systems and the livelihoods of the majority of the country's population, which are reliant on rain-fed subsistence agriculture (Swarnam et al. 2024). Climate-smart technological innovations are designed to improve resilience to variable weather patterns, reduce greenhouse gas emissions, and increase agricultural productivity and income.

In Eastern Africa—namely, Uganda, Kenya, Tanzania, Rwanda, and Burundi—the variability in staple crop yields is influenced by climate variability (Mubenga-Tshitaka et al. 2023). Improving access to technological innovations, particularly among female beneficiaries, and increasing livestock and milk production can improve the resilience of food production systems and food security in the region (Ojara et al. 2022; Warinda et al. 2020). Globally, climate change, disproportionately, threatens agricultural production in vulnerable regions with concentrated poverty and rapid population growth. This is particularly evident in Africa, Central Asia, and Latin America, where frequent droughts and variable rainfall significantly constrain agricultural output (Guo et al. 2022; Ortiz-Bobea et al. 2021).

Growing evidence suggests that livestock and milk production, facilitated by livestock donation programs, can serve as an effective climate adaptation strategy for smallholder farmers in Sub-Saharan Africa, demonstrably improving incomes, nutrition, and food security in rural communities (Argent et al. 2014; Baidoo et al. 2016; Inoni 2010; Nilsson

2019). Livestock contributes to increased incomes directly through milk and meat sales and indirectly by providing organic fertilisers that improve soil fertility and increase crop productivity (Collishaw et al. 2023; Erdaw 2023). Incorporation of livestock into smallholder systems through donation programs not only diversifies food sources but also generates alternative income streams, mitigating the risks associated with seasonal fluctuations in crop production and food availability (Banda & Tanganyika 2021; Chen et al. 2021).

The increasing prevalence of livestock donation programs as a poverty reduction strategy for smallholder farmers in low- and medium- income countries presents an opportunity to mitigate the contribution of livestock production to climate change. Leveraging livestock donation programs to promote the installation of biogas plants that utilise livestock waste to generate clean energy for domestic use can simultaneously mitigate poverty, food insecurity, and climate change (Bateki et al. 2023; Ezeanya & Kennedy 2016). Most households in rural areas of Sub-Saharan Africa rely on fuelwood as their primary energy source. However, sustained population growth and accelerating deforestation compromise the long-term viability of fuelwood as a cost-effective energy source (Shackleton et al. 2022). Supporting beneficiaries of livestock donation programs to produce biogas as a cleaner and cheaper alternative energy source for domestic use can yield economic, health, environmental, and social benefits in rural parts of Sub-Saharan Africa (Dagnachew et al. 2020; FAO 2018; Rasimphi & Tinarwo 2020). In addition, integrating biogas production as a core element of livestock donation programs could reduce the net GHG footprint of livestock donation programs (Ahmad & Jabeen 2023; Brini 2021; Dagnachew et al. 2020).

This study is aimed at investigating the net household economic effect of the Girinka program through three key research questions: (1) Do program benefits of livestock donation programs for households outweigh costs? (2) Could alternative program designs that incorporate climate-smart cowsheds and biogas plants change the potential net benefit for participants? and (3) What factors influence the potential net benefit for beneficiary households? We hypothesise that the joint distribution of biogas production plants and heifers can result in positive net benefits for rural households, contributing to improved livelihoods, economic development, and environmental sustainability. Quantifying the net benefit at the household level can provide valuable insights for policymakers and practitioners working in rural development, sustainable agriculture, and climate change mitigation.

While numerous empirical assessments have evaluated the economic and social impacts of livestock donation programs in Sub-Saharan Africa and beyond (e.g. Argent et al. 2014; Baidoo et al. 2016; Hansen et al. 2019; Inoni 2010; Kafle 2014; Rawlins et al. 2014; Salazar et al. 2018; Shikuku et al. 2017), existing evaluations primarily focus on specific outcomes such as income and nutrition improvements (Kafle 2014; Kayigema 2013; Rawlins et al. 2014) or enhanced crop productivity (Christiaensen et al. 2011; Kim et al. 2013). Although the potential economic benefits of integrating climate-smart technologies, including mitigation and adaptation options, within poverty reduction strategies like livestock donation programs are widely acknowledged (Bucagu et al. 2014; Ezeanya & Kennedy 2016; Kayigema & Rugege 2014; Klapwijk et al. 2014), thorough quantitative evaluations of such benefits remain scarce. Notable examples include Shikuku et al.'s (2017) ex-post regression analysis of climate-smart livestock technologies and Hansen et al.'s (2019) ex-post econometric study assessing the impact of these technologies on agricultural production and income.

This research contributes to the emerging body of literature on potential benefits of climate-smart agricultural practises in low- and middle-income countries (Li et al. 2023; Swarnam et al. 2024; Tabe-Ojong et al. 2024). Our evaluation framework utilises a sto-chastic household benefit–cost analysis (BCA) to explicitly quantify inherent variability in

parameter values that influence the costs and benefits of adopting climate-smart technological innovations at the household level (Akinyi et al. 2022; Mutenje et al. 2019). This approach enables the comprehensive quantification of variable household costs and benefits, accounting for inter-household variation in the cost of animal feed, water, and access to artificial insemination and veterinary services (Kayigema & Rugege 2014; Mutimura & Everson 2011). The study's unique feature lies in its comprehensive consideration of a wide range of uncertain fixed capital and variable costs and benefits from the perspective of beneficiary households. It addresses the variability in the expected per-household net benefit value due to the variability in reported unit cost and benefit values. The research evaluates the potential impact of providing biogas production plants and heifers to rural households, specifically through the *One Cow per Poor Family Program* in Rwanda's Eastern and Western provinces.

While initial evidence suggests potential challenges with ground implementation (Issahaku et al. 2024), neglecting an assessment of program modifications integrating livestock donation with biogas plant distribution could overlook opportunities for improved resource allocation within development aid programs, provided such modifications deliver a net positive economic impact. The emergence of new, affordable small-scale biogas production technologies and their successful implementation in Latin America (Garfi et al. 2016; Rocha-Meneses et al. 2023; Vásquez et al. 2024) presents a significant opportunity to consider the distribution of biogas plants as a component of livestock donation programs in Sub-Saharan Africa and other low-income countries.

The rest of the paper is organised as follows: Section 2 describes the case study area. Section 3 outlines the stochastic BCA methodology to address our key research questions. Section 4 presents the findings, while Section 5 discusses the findings in the context of the reviewed literature, along with implications, limitations, and future research directions. Finally, Section 6 draws out the key study conclusions.

2 Case study area context description

Our case study area is Rwanda's Eastern and Western provinces in Eastern Africa, one of the countries in the south of the Sahara Desert considered among the world's most food insecure regions (FAO, 2018) (Fig. 1). Due to the limited observed variation between Rwanda's Eastern and Western provinces in relation to our research objectives, we opted to analyse them as a single, combined case study area. Therefore, although geographically distinct, the Eastern and Western provinces exhibited similar characteristics relevant to our investigation, justifying their amalgamation into a unified case study area for this analysis.

Rwanda is the most densely populated country in East Africa with a population of 11.6 million and a total area of 26,338 km², 33% of which is arable land (Ezeanya and Kennedy 2016; International Fund for Agricultural Development (IFAD), 2016). Agriculture drives Rwanda's rural economy contributing substantially to food production, rural employment, and incomes. In 2015, 81% of Rwanda's population lived in rural areas, with 68% of the rural population living below the poverty line. In Rwanda, where 67% of the poor reside in rural areas and rely on agriculture, continuous growth in agricultural productivity is essential for achieving food security and reducing poverty (World Bank 2016). In 2018, 19% of households experienced food insecurity, and 38% of children under five suffer from stunted growth due to chronic undernutrition (World Bank 2018). Land use in Rwanda typically involves mixed crops, including beans, cassava, wheat, maize, and rice, along with



Fig. 1 Map of Rwanda showing the Eastern and Western provinces (shaded). Source: Locator map adapted from eMapsWorld

smallholder livestock farming systems that cover land areas ranging from 0.2 to 1 hectare (ha) per farm. The average land holding for most farmers is 0.76 ha (Nilsson 2019). Livestock farming plays a crucial role in Rwanda's agricultural production and is integral to the economic and cultural life of the country's rural areas. It is a significant source of nutrition, income, and employment, with over 70% of agricultural households engaged in livestock husbandry (Rafael 2023). The average household in Rwanda has seven to eight members and one to three cows (RGB 2018).

The average consumption of fuelwood and charcoal in Rwanda is estimated at two kilogrammes per person per day, leading to significant pressure on 16% of the country's rural land that is forested (Bikorimana et al. 2023). This high demand for fuelwood and charcoal has also increased costs, with rural households spending up to 15% of their monthly incomes on fuelwood for cooking and lighting (Anaclet 2023). Exposure to wood smoke from using fuelwood stoves in Sub-Saharan Africa has been strongly linked to respiratory diseases, particularly among women and children traditionally responsible for cooking duties (Bede-Ojimadu & Orisakwe 2020). The introduction of alternative clean energy sources for rural household energy use, with an emphasis on biogas generated from cow dung, is widely considered a logical option due to steady increases in the availability of cow dung across rural areas of Rwanda (Onyekaozuoro et al. 2023; Rubagumya et al. 2023).

The 'One Cow Per Poor Family' program, locally known as Girinka, was initiated in Rwanda in 2006 as a poverty reduction livestock donation program. Under this program, crossbred heifers are distributed to economically vulnerable households, with the requirement that the first female calf born to the recipient family is passed on to another household (Rwanda Ministry of Agriculture and Animal Resources) (MINAGRI), 2006). Households eligible for the Girinka program in Rwanda are identified by the village community based on specific criteria. These criteria include the absence of prior cow ownership, ownership of land ranging from 0.25 to 0.75 hectares, prior construction of a traditional cowshed, and classification as 'poor' according to a community-based poverty assessment system. This system assigns household vulnerability scores based on various indicators such as health, housing, food security, income, and land ownership. The primary objectives of the program are to promote increased rural milk consumption to address malnutrition issues, particularly among children.

Second, the program aims to improve household food security by increasing crop yields by adopting integrated crop-dairy farming and applying organic manure for soil fertility enhancement. Third, it aims to empower rural communities by diversifying income sources via integrated farming practises. Secondary objectives include introducing environmentally friendly agricultural production systems through emphasising zerograzing and encouraging manure utilisation through organic fertiliser production and biogas generation, offering a clean alternative to fuelwood dependence, and contributing to reduced deforestation and improved air quality.

The Girinka program is funded by the Rwandan government in partnership with the private sector, civil society organisations, local non-government institutions, and international organisations. Between 2006 and 2015, the Girinka program distributed 297,060 heifers to 297,060 rural households impacting over 1.2 million individuals, representing around 16% of Rwanda's total rural population (Rwanda Governance Board (RGB), 2018). A more recent empirical investigation has examined the progression of the Girinka program's implementation, coverage, and effectiveness, affirming the assertions made by the RGB (Habiyaremye et al. 2021). Figure 2 shows how cows were distributed between 2006 and 2015 across Rwanda's five provinces under the Girinka program. The Rwandan government intends to reach more than 700,000 poor households by 2035 under the Girinka program.

While the Girinka program has been primarily associated with positive impacts, such as increased agricultural production and household income (Argent et al. 2014; Nilsson et al. 2019), it has been observed that the program can impose an energy cost burden on rural households (Khundi-Mkomba et al. 2023). Limited access to veterinary services and adequate water supply imposes substantial financial burdens on program beneficiaries, potentially undermining the economic viability of the intervention for beneficiary households (Sapp et al. 2023).



Fig. 2 Distribution of cows between 2006 and 2015 (left) and across Rwanda's five provinces (right). Source: Adapted from RGB (2018, p. 13)

While biogas technology can potentially improve rural livelihoods and reduce energy costs for low-income Rwandan households, its adoption has been remarkably low due to prohibitive capital set-up costs and high dis-adoption rates (Lwiza et al. 2017; Muke-shimana et al. 2021). The Rwandan government put in place energy policies to prioritise biogas energy production and use in recognition of the opportunity presented by the Girinka program to reduce deforestation and GHG emissions (Ezeanya and Kennedy 2016). A government subsidy program aimed at providing materials and technical support to rural households in Rwanda's Eastern Province to incentivise biogas generation at the household level has not yet proven to be effective (Ezeanya & Kennedy 2017; Roopnarain and Adeleke 2017).

This research is motivated by the widespread discussion in the literature about the potential for realising positive economic, health, social, and environmental externalities from the Girinka program through the adoption of biogas production (Berhe et al. 2017; Bucagu et al. 2014; Geddafa et al. 2023; Kayigema & Rugege 2014; Klapwijk et al. 2014). However, the potential net benefit at the household level is rarely quantified. This study addresses this gap by quantitatively assessing the potential net benefits of adopting biogas production for domestic energy use among Girinka beneficiary households. Given the potential influence of implementation challenges on the current prevalence and efficiency of biogas plants (Kalina et al. 2022), we discuss policy tools that could address these obstacles and potentially contribute to more widespread and effective adoption.

3 Methodology

A six-step ex-ante benefit–cost analysis (BCA) was conducted to estimate present values (PV) of costs, benefits, and net benefits under the current program design and three alternative program design scenarios from the perspective of beneficiary households (Fig. 3).

The baseline scenario, or 'without project' scenario, served as a reference point for identifying and quantifying additional costs and benefits of the current program and the alternative program designs. The counterfactual 'without project' scenario did not involve a control group as commonly seen in experimental designs, noting that this was not an impact evaluation, but rather a conceptual construct that explores potential future trajectories under the assumption of the project not being implemented. The analysis involved identifying relevant costs and benefits, calculating net benefits, and comparing incremental costs with additional benefits of switching to alternative program design scenarios. The costs and price values were converted to 2020 USD equivalent values to standardise the cost and benefits used in net benefit calculations. The net PV benefit to households was calculated over 25 years between 2018 and 2043, and discount rates between four and seven percent were used. The analysis also included a sensitivity analysis to test the robustness of household net benefit estimates. Cost and price values were adjusted for inflation using US government CPI data. The following sections provide further details describing how each of the six steps was implemented.

3.1 Scenario description

Table S.1 in Supplementary Material presents an overview and the assumptions considered for the baseline, current program, and three alternative program design scenarios, including program costs and household costs and benefits.



Fig. 3 Flowchart depicting the steps involved in benefit-cost analysis of integrating climate-smart technological innovations to Rwanda's livestock donation program

Our counterfactual baseline (without project) depicts a scenario with few households owning heifers and traditional cowsheds, and climate-smart technologies, including improved cowsheds and biogas plants, are virtually absent. Subsistence rainfed cropping systems predominate. Climate-smart cowsheds are equipped with effective rainwater harvesting, flooring, and waste management systems to ensure efficient removal and storage of manure and disease management. As such, climate-smart cowsheds experience less manure production loss and milk loss due to mastitis than traditional cowsheds made from locally found materials, with basic flooring and no storm and wastewater management system (IFAD 2016). Mastitis infections in dairy cows are primarily caused by poor hygiene practises, including ineffective cowshed waste management, which can lead to udder infections and a reduction in milk yield and quality (Iraguha et al. 2015). Additionally, in our sensitivity analyses, we quantified the net benefit value under various baseline scenarios to account for different initial adoptions of various combinations of heifers, climate-smart cowsheds, and biogas plants prior to program intervention, in order to adequately address varying levels of asset ownership across beneficiary households (Robinson & Hammitt 2017).

In the initial analysis, the baseline scenario served as a reference point to estimate the net benefit to households under the current Girinka program design, which primarily focused on providing one lactating heifer per poor rural household, along with the requirement for beneficiaries to construct a traditional cowshed. Subsequently, the net benefit of transitioning from the baseline scenario to a scenario involving the distribution of climatesmart cowsheds to beneficiary households in addition to lactating heifers was calculated. We further assessed the net benefit of adding affordable biogas production plants (tubular polyethylene biodigesters) to the traditional heifer and cowshed package. This involved accounting for the cost savings households would accrue by replacing fuelwood, their primary domestic energy source, with biogas. Moving beyond traditional program elements, we investigated the 'everything scenario', where households received all interventions: heifers, biogas plants, and climate-smart cowsheds. Net benefits were then estimated relative to the baseline for comparison. To assess the robustness of our findings, we conducted further sensitivity analyses to evaluate net benefits under various baseline assumptions.

3.2 Identifying benefits and costs

Informed by a review of literature on livestock donation programs and household biogas production, we identified costs and benefits for beneficiary households (Table 1).

The BCA framework comprises three primary cost and benefit components, namely: (1) fixed capital set-up costs, (2) variable costs, and (3) household benefits (Fig. 4).

The costs were broadly categorised into fixed capital set-up costs (cowshed construction costs) and variable operations and maintenance (O&M) costs (feed, watering, artificial insemination, and veterinary services and labour). The direct benefits of owning a heifer include manure, calves, milk, and meat consumption and revenue. Owning a biogas plant can lead to health benefits and reduced GHG emissions and reduced deforestation. Health, social, cultural, and environmental benefits are rarely quantified. The framework incorporated three main categories of fixed program costs: (1) capital set-up for lactating heifers, including purchase and distribution overhead with transaction costs; (2) augmentation of traditional cowsheds with climate-smart features like rainwater harvesting, flooring, and waste management; and (3) installation of biogas production plants. To align with the Girinka program's practise of paying a fixed fee per delivery, independent of the number of

Costs and benefits	Finding		
Household costs			
Capital set-up costs			
Cowshed construction costs	Households spend between USD 31 and USD 62 per cow on cowshed installation per year depending on the quality of materials used (Kayigema 2013). Top end climate-smart cattle shelter can cost up to USD165 per cow (Miklyaev et al. 2017). A climate-smart cowshed includes a rainwater harvesting system, good quality flooring, and a waste management system for efficient removal and storage of dung and urine and improved feed and watering points (IFAD 2016)		
Operation and maintenance costs			
Feed	The cost of feeding a heifer varies significantly depending on the type of feed used and can range between USD 119 and USD 2837 per cow (IFRC 2016; Miklyaev et al. 2017)		
Watering	Households spend between USD 15 and USD 730 per cow per year on watering (IFAD 2016; Kayigema 2013; Mikly- aev et al. 2017)		
Artificial insemination	The Rwandan government provides artificial insemina- tion services at a heavily subsidised cost to smallholder dairy farmers (IFAD, 2016). Use of artificial insemination services is low, with 58% of farmers having access artificial insemination services (USAID 2016)		
Veterinary services	Treating against risk of mastitis can incur a cost of up to USD26 per cow in veterinary services, and USD12 per cow for treatment (Mwabonimana et al. 2015). Annual veteri- nary costs can total between USD 61 and USD 70 per cow (Miklyaev et al. 2017)		
Labour	The cost of labour to care for animals can range between USD 47 and USD 79 per cow per year depending on the quality of care provided (Miklyaev et al. 2017). Family labour is usually sufficient, with average household popula- tion of eight people per household (RGB 2018)		
Household benefits			
Benefits of owning a heifer			
Manure	Over 90% of Girinka beneficiaries use manure, and report increased yields and improved soil fertility due to manure use (Kim et al. 2013, 2011)		
Calves	Girinka beneficiaries reported selling the second calf for up to USD 329 per calf (IFRC 2016). The average calving interval is typically between 15 and 18 months (Miklyaev et al. 2017)		
Milk consumption and revenue	Provision of heifers with training increased milk produc- tion for household consumption and for sale (Argent et al. 2014). Girinka contributed 89% increase in milk production between 201and 2015 (RGB 2018)		

 Table 1
 Summary of benefits and costs of owning a heifer and a biogas production plant to households and basic assumptions based on a review of literature

24

Costs and benefits	Finding		
Meat consumption and revenue	Meat sales from post-lactation cows that have been culled can generate significant income for households (Salazar et al. 2018). Rawlins et al. (2014) found substantial impacts of cow transfers on household meat consumption and children's nutrition outcomes. The average calving interval was assumed as ranging between 15 and 18 months, and after four intervals, a heifer is culled and slaughtered to be consumed or sold for meat (Miklyaev et al. 2017)		
Benefits of owning a biogas plant			
Household energy cost savings	 Biogas use for household energy needs can save households from incurring monthly energy expenditures of up to USD 15 per household (Mwakaje, 2008; Surendra et al., 2014). Integrating biogas to Girinka program can yield significant energy cost savings for rural households (Ezeanya & Kennedy 2017) 		
Health benefit from reduced pollution	Fuelwood powered stoves used for cooking emit toxic gases linked with high prevalence of respiratory diseases (Nix et al., 2022)		
Reducing GHG emissions and deforestation	Widespread adoption of biogas can reduce GHG emissions and deforestation (Garfí et al., 2016; Katuwal and Bohara, 2009; Paul et al., 2017)		

Net Economic Benefit



Fig. 4 Organisational structure used to estimate costs and benefits of current and alternative Girinka design programs

cows (International Federation of Red Crescent Societies (IFRC), 2016), transaction costs were treated as fixed in this study. This assumption simplifies the model while representing the program's actual cost structure. Notably, fixed capital set-up for individual households factored in the cost of a standard cowshed lacking storm and wastewater management, reflecting current Girinka program requirements.

Variable costs encompassed ongoing O&M expenses for feed, water, artificial insemination (AI), and veterinary services. Household benefits included both consumption and revenue streams from milk, calves, and post-lactation cow sales. While potential yield increases from increased manure application in subsistence farming (Kim et al. 2013, 2011) were acknowledged, they were not quantified due to a potential trade-off with labour allocation for heifer husbandry, expected to decrease crop yields.

3.3 Calculating costs and benefits

This section illustrates the calculation of individual component costs and benefits, ultimately contributing to the household net benefit estimation. Parameter descriptions, notations, units, value ranges, underlying assumptions, and source references for the subsequent mathematical equations are provided in Table 1 and S.3 in Supplementary Material.

3.3.1 Fixed capital set-up costs

In the 'without project' scenario, households bore the full cost of acquiring a lactating heifer. Fixed program costs during the initial year encompassed heifer procurement and distribution, including transaction costs. These transaction costs covered delivery logistics, with maintenance during transit (hiring and operating facilities, staff for transport, feeding, watering, cleaning, and cow dung disposal), and institutional overheads (hiring program leaders, focus group workshops for beneficiary identification and validation, basic animal husbandry training, and cowshed construction support) (IFAD 2016; IFRC 2016).

The PV of fixed costs of constructing a traditional cowshed, PV_Constr_{*T*trad}, was calculated as the sum of the initial construction's PV in year t_0 and the PV of its replacements at the end of its lifespan throughout the 25-year analysis period. This can be expressed as:

$$PV_Constr_{T_{trad}} = \sum_{t_{trad}} \frac{Constr_{trad}}{(1 + DR)^{trad}}$$

for $t_{trad} = 0, 4, 8, 12, 16, 20$ (1)

where t_{trad} is the year that a replacement traditional cowshed is constructed every five years after the initial construction year, t_0 (i.e. $t_{\text{trad}}=0$ inclusive) with the average lifespan of a traditional cowshed estimated at 5 years. DR is the discount rate.

The PV of fixed capital costs of constructing climate-smart cowsheds with a longer lifespan, estimated at 12 years (PV_Cowshed_{Tcs}), were calculated as:

$$PV_Cowshed_{T_{cs}} = \sum_{t_{cs}} \frac{\text{Dist}_{cs} + \text{Constr}_{cs}}{(1 + \text{DR})^{t_{cs}}}$$
for $t_{cs} = 0, 11, 22$
(2)

where Dist_{cs} is the cost of distribution, Constr_{cs} is the cost of construction, and t_{cs} denotes the year of replacement for a climate-smart cowshed. This occurs every 11 years, starting from the initial year of program implementation, t_0 , considering the average lifespan of 11 years. The PV cost of constructing a tubular polyethylene biodigesters (PV_Biod_{*Tbiogas*}) considered the most cost-effective biogas technology based on Garfi et al. (2016) was calculated as the sum of two components: (1) the PV of initial distribution and construction costs incurred in the program's first year, and (2) the PV of periodic replacements throughout the 25-year analysis period. Each replacement cost, discounted to its PV, was factored in for the years coinciding with the end of the biodigester's five-year lifespan, t_{biogas} :

$$PV_Biod_{T_{biogas}} = \sum_{t_{biogas}} \frac{\text{Dist}_{biogas} + \text{Constr_{biogas}}}{(1+DR)^{t_{biogas}}}$$

for $t_{biogas} = 0, 4, 8, 12, 16, 20$ (3)

where $\text{Dist}_{\text{biogas}}$ is the fixed capital costs of distributing biogas plants, and $\text{Constr}_{\text{biogas}}$ denotes biogas plant construction costs comprised of two main components: the acquisition of all necessary materials and the fees associated with soliciting technical support to supervise the installation process, as recommended by Garfi et al. (2016).

For each scenario, i, the total fixed cost (PFC_i) per household was calculated as the sum of the cost of a lactating heifer (Heif), transactions costs (Trans), cost of a cowshed (Cowshed), and the cost of a biodigester (Biod):

$$PFC_i = \alpha(Heif) + \beta(Trans) + \sigma(Cowshed) + \gamma(Biod)$$
(4)

where $\alpha = 1$ and $\beta = \gamma = 0$ under the baseline scenario with the household incurring the cost of a lactating heifer. $A = \beta = \sigma = 1$ and $\gamma = 0$ under the current Girinka program and under the scenario that incorporates distribution of climate-smart cowsheds under the current program design with the household incurring the cost of constructing a traditional cowshed. $A = \beta = \sigma = \gamma = 1$ under scenarios that incorporate distribution of biogas plants with the household incurring the cost of constructing a traditional cow-

3.3.2 Variable costs

The PV of total household O&M costs, PVC₁, was calculated by aggregating the PVs of individual cost components: feed, water, artificial insemination, and veterinary services:

$$PVC_{I} = \sum_{i} \frac{O\&M_{i}}{(1+DR)^{T}}$$

for *i* = feed, water, AI, veterinary (5)

where $O\&M_i$ is the annual operations and maintenance cost. In the initial year of program implementation, t_0 , artificial insemination costs were assumed to be zero, reflecting the current program's common practise of distributing lactating (pregnant) cows, negating the immediate need for artificial insemination services.

3.3.3 Household benefits

Household benefits were estimated as the present value of revenues from milk sales (including the imputed value of household consumption), m, calf sales, k, and post-lactation cow sales, p. Additionally, scenarios featuring biogas production plant distribution incorporated the PV of energy cost savings as a further benefit.

The PV of revenues from milk sales, PV_Rev_{*Tm*}, was calculated as:

$$PV_Rev_{T_m} = \sum_{j} \frac{\frac{Prod_m \times \Delta P_waste_j \times Price_m}{(1+DR)^T}}{(1+DR)^T}$$
 (6)
for j = traditional cowshed, climatesmart cowshed

where Prod_m is the total volume of milk produced, ΔP _waste is production loss rate, and Price_m is the market price of milk. Higher mastitis risk in traditional cowsheds (Iraguha et al. 2015; Juozaitiene et al. 2006) led to greater milk loss compared to climate-smart designs. Notably, milk losses under climate-smart cowsheds were primarily attributed to the lack of milk storage facilities (IFAD 2016).

The value of the benefit of producing calves was calculated as the sum of the PV of revenues from sale of the thirdborn calf, PV_Rev_{Tk} . Consistent with Girinka program requirements, we assumed households gifted the firstborn calf, retained the second for herd expansion, and sold the third as an income source (Miklyaev et al. 2017). Fourthborn calves were kept as replacements for culled cows. Given a heifer's 3–5-year lactation cycle and 1–1.5-year calving interval, only the third calf, born after seven years (three intervals), entered the revenue calculation (IFRC 2016).

$$PV_Rev_{T_k} = \sum_{t_k} \frac{Price_k}{(1+DR)^{t_k}}$$

for $t_k = 7$ (7)

where $Price_k$ is the market price of a calf.

The PV of revenue from post-lactation cow sales, PV_Rev_{Tp} , was determined by summing the PV of two components: (1) the initial sale of the first cow at the end of its nine-year lactation cycle (eight years after program implementation) and (2) subsequent sales of replacement cows at nine-year intervals throughout the 25-year analysis period:

$$PV_Rev_{T_p} = \sum_{t_p} \frac{Price_p \times D}{(1+DR)^{t_p}}$$

for $t_p = 8, 16, 24$ (8)

where Price_p is the market price of a cow. The analysis assumes a potential depreciation in market value, *D*, reflected in lower sale prices, for culled cows due to anticipated diminished meat quality (International Finance Corporation (IFC), 2007; IFRC 2016).

The PV of household energy cost, PV_Energy_i , was estimated by quantifying the financial benefit of substituting biogas for expensive traditional energy sources, primarily fuelwood, used predominantly for cooking:

$$PV_Energy_{j} = \sum_{j} \frac{(EC) \times Util_{j}}{(1+DR)^{T}}$$
(9)

for j = traditional cowshed, climate smart cowshed

where EC is the cost of energy and Util_j is the utilisation rate of biogas plants. Higher biogas plant utilisation was projected for climate-smart cowsheds compared to traditional designs due to enhanced manure management efficiencies. Climate-smart structures facilitate effective manure collection and storage, minimising losses and maximising biogas production potential and biogas plant utilisation rate. Consequently, energy cost savings, calculated as the difference in energy costs under traditional and climate smart cowsheds, were only factored into scenarios involving biogas plant distribution.

For each scenario, *i*, the total PV benefit (PVB_i) per household was calculated as:

$$PVB_{i} = PV_Rev_{T_{u}} + PV_Rev_{T_{u}} + PV_Rev_{T_{u}} + \lambda(PV_Energy_{i})$$
(10)

where $\lambda = 1$ for scenarios involving biogas plant distribution, and $\lambda = 0$ for all the other scenarios.

The net benefit per household was calculated as the difference between total household benefits and costs. To capture inherent variability in parameters, input values for these calculations were drawn from pre-defined ranges. A comprehensive description of this parameter value selection process, designed to adequately quantify uncertainty, follows in the subsequent section.

3.4 Data

This study utilised data from both existing primary survey data and local studies (peerreviewed publications and consulting reports). This study leverages data collected from a 2018 cross-sectional survey of Rwandan Girinka program participants conducted by our co-authors, focusing specifically on household financial costs and benefits, to further explore the program's economic impact. Conducted by the University of Rwanda's College of Agriculture, Animal Sciences and Veterinary Medicine, the survey aimed to understand the program's impact on household costs, income, and food and nutrition security. For a detailed description of the survey design and implementation procedures, please refer to the Supplementary material.

Parameter values were drawn from both survey data and local studies (peer-reviewed publications and consulting reports). Instead of relying solely on averages or medians, value ranges and probability density functions (PDFs) were derived for key parameters. This approach incorporated variability observed in survey responses and peer-reviewed literature (IFAD 2016; Kayigema & Rugege 2014; Mutimura & Everson 2011). Validation by a local expert with field experience at the Rwanda Agriculture Board, in consultation with program coordinators, ensured alignment with real-world conditions. For example, ranges for animal feed costs, water, artificial insemination, veterinary services, and milk prices were adjusted to reflect estimates from similar studies. A summary of parameter descriptions, notations, units, ranges, and data sources is provided in Supplementary Material (Table S.3). Values in Table S.3 were converted to per year per household equivalents to facilitate subsequent calculations of present value costs, benefits, and net benefits for each household. Additionally, sensitivity analyses were conducted to explore the sensitivity of the results of our analysis under varying assumptions about respondents, further strengthening the confidence in our conclusions.

3.5 Quantifying variability in parameter values

We quantified the variability in key parameters influencing net benefit calculations by fitting probability density functions (PDFs) to observational data from the survey and secondary data from the literature. Various functional forms, including exponential, log-logistic, and lognormal, were employed to model the variability in parameters such as income, feeding costs, water costs, artificial insemination costs, veterinary service costs, milk consumption, and revenue values (Fig. 5). The optimal PDF for each parameter was selected based on chi-squared goodness-of-fit tests.

For all other parameters, variability was modelled using the beta distribution. This continuous PDF resembles a truncated normal distribution, offering a symmetrical bell-shaped



Fig. 5 Fitting probability density functions (red line) to frequency distributions from cross-sectional 2018 survey data from Eastern and Western Rwandan provinces

curve centred around the median value and confined by the known range. The beta distribution is well-suited for representing uncertainty in parameters with known medians and ranges, making it ideal for incorporating data gathered from the reviewed literature of the case study region. Further technical details regarding selecting appropriate PDFs for uncertainty quantification in net benefit calculations can be found in Kandulu and Connor (2017). We calculated Pearson correlation coefficients using survey data to assess potential correlations between milk production and key cost/revenue factors. This analysis examined the relationships between milk production and the costs of feed, water, and veterinary services, as well as the relationship between milk production and milk price.

We employed stochastic Monte Carlo simulations to simulate the inherent variability in key parameters and understand its impact on net benefit. This approach involved iteratively drawing random parameter values from the pre-defined PDFs. These parameter values were then implemented in the established net benefit equations, resulting in 1000 unique net benefit calculations per scenario. The simulations incorporated correlations between certain parameters, ensuring realistic variability patterns. The resulting 1000 net benefit values for each scenario were then used to generate frequency distributions, characterising the potential range and distribution of net benefits under each simulated condition.

3.6 Sensitivity analysis

To explore the impact of existing asset ownership on the projected benefits of the Girinka program, we conducted a sensitivity analysis by examining net benefits under three alternative baseline scenarios. These scenarios varied the initial levels of adoption of key program components—heifers, climate-smart cowsheds, and biogas plants—among the reference households prior to program intervention. Our reference household practised subsistence cropping and did not own a heifer, reflecting the target population of the program with no prior livestock ownership. An alternative baseline scenario introduced additional complexity by assuming the reference household already owned a heifer, but still received a climate-smart cowshed and biogas plant. This allowed isolation of the net benefit contributions of these interventions beyond heifer ownership. In a third 'partial adoption' baseline scenario, the reference household already owned a heifer and a biogas plant, but received a climate-smart cowshed. This analysis focused on the incremental benefit of the cowshed in a context where other program components were already present. By examining net benefits across these diverse baseline scenarios, a more nuanced understanding of the program's potential impact under different existing asset ownership conditions is provided.

4 Results

Table 2 presents present value (PV) costs, benefits, net benefits, and benefit–cost ratios for households under the existing Girinka program design and three alternative program scenarios.

Without the Girinka program, even the few resource-constrained rural households who could afford a heifer would face a PV cost of USD 1660. However, these households could also anticipate an average PV benefit of USD 3050, resulting in a net benefit of USD 1390 per household on their investment. The current Girinka program delivers an average net benefit of USD 2277 per beneficiary household, translating to a benefit–cost ratio (BCR) of 4:1. Incorporating climate-smart cowsheds and biogas production plants into the program design could generate an even higher BCR of 5:1.

Scenario	Program cost	Household cost	Total household benefit	Net household benefit	Household benefit-cost ratio
Current Girinka program	910	763	3040	2277	4.0
Girinka+climate-smart cowsheds	957	805	3280	2475	4.1
Girinka+biogas plants	3792	1185	5425	4271	4.6
Girinka + climate-smart cow- sheds + biogas	3839	1228	6157	4929	5.0

 Table 2
 Expected present value household costs, benefits, net benefits (USD), and benefit–cost ratio estimates under alternative program designs

Implementing climate-smart cowsheds and biogas production plants alongside heifer distribution in the Girinka program would incur an additional program cost of USD 2929 per household (USD 910 baseline cost increasing to USD 3839). This scenario would yield a net benefit of USD 3117 per household (from USD 6157 to USD 9274, USD 1727 higher than the base scenario). Introducing biogas plants alone, at an additional cost of USD 2,882 per household, generates a net benefit of USD 2385 per household.

Our sensitivity analysis considered the incremental net benefit under three alternative baseline scenarios. Under the first alternative baseline scenario where our benchmark household practised subsistence rainfed cropping and did not own a heifer, receiving a lactating heifer, climate-smart cowshed, and biogas plant generated a net benefit of USD 4929 per household and a BCR of 5:1. In an alternative analysis where our reference household already owned a heifer, the net benefit from receiving a climate-smart cowshed and biogas plant decreased to USD 2652 per household, with a BCR of 7:1. For households already owning both a heifer and a biogas plant, the net benefit from adding a climate-smart cowshed decreased to USD 689 per household.

Figure S.1 in the Supplementary Material presents the frequency distributions and summary statistics of net benefit values calculated using 1000 random samples drawn from the PDFs of variable cost and benefit parameters. For the baseline scenario where households only incur the cost of a lactating heifer, net benefit estimates were predominantly positive, with minimal probability of net losses. However, net benefit values varied substantially, with standard deviations ranging from 29 to 50% of the expected value.

Figure S.2 in the Supplementary Material presents tornado graphs quantifying the sensitivity of net benefit calculations to various parameter values used to estimate costs and benefits under each scenario. The analyses reveal that milk price and production consistently contributed the most to variability in net benefit estimates across all scenarios. Importantly, even for these influential parameters, variability within their probable ranges did not significantly alter the key conclusion: households experience a substantial net benefit increase when the current Girinka program design is augmented with climate-smart cowsheds and biogas production plants. This is illustrated, for example, by varying milk prices (the most sensitive parameter) across its entire range while holding other parameters at their medians. This manipulation only caused net benefit estimates to range between USD 1229 and USD 2988 under the current program and between USD 3617 and USD 5942 under the scenario with climate-smart technologies.

In addition, our analysis revealed positive correlations between key inputs and milk production in the Girinka program, mirroring existing literature. Consistent with Gonzáles et al (2016), the correlation coefficient between animal feed expenditure and milk production was calculated at 0.29. Similarly, the correlation between water consumption and milk productivity aligned with Kayigema (2013) at 0.30. Additionally, the calculated correlation between veterinary visits and milk production of 0.21 resonated with Argent et al. (2014). These findings provide empirical support for the importance of these factors in influencing milk production outcomes for beneficiary households.

5 Discussion

This study uses a case study in Rwanda's Eastern and Western provinces to evaluate the economic performance of the current Girinka program and three alternative designs. These alternative designs focus on incorporating the distribution of climate-smart

cowsheds and biogas plants alongside the program's current core element: distributing heifers to low-income households. To assess economic viability, we utilise a BCA framework to account for the inherent variability in key parameters affecting costs and benefits. The variability reflects the disparate circumstances of beneficiary households, particularly regarding the cost of animal feed, water, artificial insemination, and veterinary services.

Our analysis reveals that the Girinka program consistently delivers positive, albeit highly variable, net benefits to beneficiary households. While introducing climate-smart cowsheds alone alongside, the current heifer distribution offers a limited additional economic advantage, a combined package of climate-smart cowsheds and biogas production plants significantly enhances household net benefits. While the current Girinka program already delivers a substantial average net benefit of USD 2277 per household (BCR: 4:1), incorporating climate-smart cowsheds and biogas production plants presents even greater potential. This enhanced program design could yield a BCR of 5:1, suggesting significant benefits for beneficiary households. The findings of our study are consistent with a growing body of literature indicating the positive net benefits of incorporating climate-smart technological innovations into livestock donation programs, although these benefits have not been quantified (Berhe et al. 2017; Bucagu et al. 2014; Geddafa et al. 2023; Kayigema & Rugege 2014; Klapwijk et al. 2014). Estimates of net benefits exhibit sensitivity to fluctuations in milk production and prices, suggesting that substantial increases in milk supply could potentially diminish net benefits through price reductions. The differences in milk production among households may be attributed to varying cow-feeding intensities, which in turn are influenced by the affordability of food supplements such as commercial feed and vitamins (Manzi et al. 2020; Wilkes et al. 2020; Wright et al. 2016). Despite documented positive impacts like increased agricultural production and household income (Argent et al. 2014; Nilsson et al. 2019), the Girinka program has also revealed potential drawbacks. Khundi-Mkomba et al. (2023) suggest that limited access to veterinary services and water supply can impose significant financial burdens on beneficiary households, potentially negating the program's economic viability (Sapp et al. 2023).

This study exclusively focused on the economic benefits of incorporating biogas production into the Girinka program. However, a comprehensive BCR assessment would necessitate quantifying the additional health and social benefits accrued by households. These include reduced reliance on fuelwood, leading to potential decreases in respiratory infections, time and effort spent fetching fuelwood, and domestic chores traditionally carried out by women and children (e.g. fire preparation, kitchen cleaning, and dishwashing) (Njenga et al. 2023; Sepee & Tesfahun 2023). Furthermore, incorporating the environmental benefits of reduced deforestation and greenhouse gas emissions associated with biogas adoption would likely further enhance BCR estimates.

Our findings support the growing calls to harness the broader environmental and social benefits achievable through integrating biogas plant installations to livestock donation programs (Ahmad & Jabeen 2023; Khundi-Mkomba et al. 2023; Onyekaozuoro et al. 2023). Our study demonstrates that community-scale interventions targeted at low-income households can yield high BCRs, delivering direct, quantifiable benefits directly attributable to the interventions. This work gains relevance in light of recent advancements in affordable small-scale renewable energy technologies (Clemens et al. 2018; Gitau et al. 2019; Jagger & Das 2018; Keerthana Devi et al. 2022). Future research can build upon this foundation by quantifying the GHG emissions and health benefits associated with climate-smart technology adoption. In addition, non-market valuation techniques can be employed to quantify social impacts to broaden the scope of quantified costs and benefits. While awareness of fuelwood's negative health and environmental impacts grows, and cleaner alternatives become increasingly available, its dominant use for cooking persists in low- and middle-income countries. This can be partly explained by disparities in the perceived value of male and female labour within households, impacting the adoption of innovations with potential net benefits (Overfield 1998). Specifically, fuelwood collection often falls to women, and their limited bargaining power in household decision-making can hinder the adoption of cleaner options. This is supported by Behera and Ali (2016), who found female-headed households in Sub-Saharan Africa to be more likely to adopt cleaner energy sources and less reliant on fuelwood. Consequently, parallel programs promoting women's financial empowerment through improved microcredit access could contribute to increased adoption of cleaner technologies by enhancing their bargaining power within the household.

Recent evidence suggests potential challenges with ground implementation of small-scale biogas digesters, which may hinder their adoption (Issahaku et al. 2024; Kalina et al. 2022). These challenges include high capital and ongoing O&M costs, poor monitoring and maintenance of existing digesters, and failures shortly after installation negatively. Recognising the potential obstacles associated with implementing climate-smart technologies like biogas plants and climatesmart cowsheds, we recommend a multi-faceted policy approach to overcome barriers hindering the widespread adoption of climate-smart technologies like biogas plants and climate-smart cowsheds. This includes (1) cost reduction through technological innovation, locally adapted models, and utilisation of cheaper materials; (2) implementation of microfinance schemes with low-interest rates and flexible payback periods to address financial constraints; (3) investment in education and training programs to empower the private sector and ensure quality construction; and (4) active participation of women, as primary users and beneficiaries, in decision-making, training, and maintenance to promote skill development and positive social acceptance. Multifaceted policy approaches tailored to local contexts and encompassing a range of adoption barriers are demonstrably more effective in promoting the uptake of climate-smart technologies than simplistic, single-objective interventions (Mukeshimana et al. 2021; Ogisi & Begho 2023). Future research should explore the impact of cost-sharing arrangements between government/ aid agencies and households on net benefits. Evaluating various cost-sharing models could offer valuable insights into program affordability and long-term sustainability.

6 Conclusion and policy implications

6.1 Conclusions

This study quantifies the net benefits of incorporating climate-smart cowsheds and biogas plants into the Girinka livestock donation program in Rwanda, demonstrating the potential for these technologies to enhance economic viability, foster environmental sustainability, and deliver direct, quantifiable benefits. Our study finds that integrating climate-smart cowsheds and biogas production plants along with heifer distribution can significantly improve the economic viability of the Girinka program. Households owning cows under this improved program design can realise net benefits 3.5 times higher than under the current design, with BCRs reaching 5:1. This finding underscores the need to integrate climate-smart technologies into agricultural development programs in low-income countries, particularly for resource-constrained smallholder communities. While variability in milk production and prices is an important consideration, the sub-stantial net benefit achieved through a combined package of climate-smart cowsheds and biogas

plants surpasses traditional program designs and highlights the superiority of multi-dimensional interventions in rural development. Incorporating non-market valuation techniques into future research, such as willingness-to-pay surveys to estimate the value of improved health outcomes and life cycle assessments to quantify greenhouse gas emission reductions, would enhance the comprehensiveness of the program's evaluation and inform evidence-based policy decisions.

6.2 Policy implications

Multi-faceted policy packages, tailored to address the diverse and localised barriers specific to climate-smart technologies, can achieve higher adoption rates than interventions with singular objectives. The combined implementation of (1) training and education programs for households to build their capacity in the maintenance and effective utilisation of biogas plants; (2) cost reduction strategies, such as pursuing technological innovation for cheaper solutions and utilising affordable local materials to decrease prohibitive upfront capital costs; (3) microfinance schemes with flexible payback periods and low-interest rates to enhance affordability and facilitate uptake; and (4) education and training programs for the private sector to ensure quality construction and installation further promoting high adoption rates. In addition, actively involving women in training, maintenance, and decision-making processes to promote skill development, encourage positive social acceptance, and acknowledge the predominant role women play in domestic energy decisions, such as firewood collection can increase uptake. This can be further reinforced by parallel programs to enhance women's financial empowerment through improved microcredit access. By simultaneously addressing multiple adoption barriers through such multifaceted and context-specific policy packages, the uptake of climate-smart technologies like biogas plants can be significantly increased compared to single-objective interventions.

Supplementary Information The online version contains supplementary material available at https://doi.org/10.1007/s11027-024-10120-w.

Funding Open Access funding enabled and organized by CAUL and its Member Institutions. This study was supported by an Australian Postgraduate Student Award, and Australian Research Council FT140100773.

Data availability The datasets generated during and/or analysed during the current study are available from the corresponding author on reasonable request.

Declarations

Conflict of interest The authors declare no competing interests.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Abebe F, Zuo A, Wheeler SA, Bjornlund H, Chilundo M, Kissoly L, Dube T (2022) The influences on farmers' planned and actual farm adaptation decisions: evidence from small-scale irrigation schemes in South-Eastern Africa. Ecol Econ 202:107594. https://doi.org/10.1016/j.ecolecon.2022.107594
- Abebe F, Wheeler SA, Zuo A, Bjornlund H (2023) The welfare enhancing effects of agricultural innovation platforms and soil monitoring tools on farming household outcomes in southeastern Africa. Int J Agric Sustain 21(1):2184586. https://doi.org/10.1080/14735903.2023.2184586
- Adesete AA, Olanubi OE, Dauda RO (2023) Climate change and food security in selected Sub-Saharan African Countries. Environ Dev Sustain 25(12):14623–14641. https://doi.org/10.1007/ s10668-022-02681-0
- Ahmad M, Jabeen G (2023) Biogas technology adoption and household welfare perspectives for sustainable development. Energy Policy 181:113728. https://doi.org/10.1016/j.enpol.2023.113728
- Akinyi DP, Ng'ang'a SK, Ngigi M, Mathenge M, Girvetz E (2022) Cost-benefit analysis of prioritized climate-smart agricultural practices among smallholder farmers: evidence from selected value chains across sub-Saharan Africa. Heliyon 8(4):e09228. https://doi.org/10.1016/j.heliyon.2022. e09228
- Anaclet N (2023) Evaluation of program decentralization of the small-scale energy systems construction and maintenance management in Rwanda: a case study of the Rwanda Domestic Biogas Program [Masters Thesis, Jomo Kenyatta University]. Retrieved 27 January 2024 from ir.jkuat.ac.ke/handle/123456789/6019
- Argent J, Augsburg B, Rasul I (2014) Livestock asset transfers with and without training: evidence from Rwanda. J Econ Behav Organ 108:19–39. https://doi.org/10.1016/j.jebo.2014.07.008
- Austin KG, Beach RH, Lapidus D, Salem ME, Taylor NJ, Knudsen M, Ujeneza N (2020) Impacts of climate change on the potential productivity of eleven staple crops in Rwanda. Sustainability 12(10). https://doi.org/10.3390/su12104116
- Baidoo S, Yusif H, Anwar U (2016) The effect of smallholder livestock production on income of farm households in Northern Ghana. J Sci Technol 36:8–19. https://doi.org/10.4314/just.v36i3.2
- Banda LJ, Tanganyika J (2021) Livestock provide more than food in smallholder production systems of developing countries. Anim Front 11(2):7–14. https://doi.org/10.1093/af/vfab001
- Baptista DM, Farid M, Fayad D, Kemoe L, Lanci LS, Mitra P, Muehlschlegel TS, Okou C, Spray JA, Tuitoek K, Unsal FD (2022) Climate change and chronic food insecurity in Sub-Saharan Africa. Departmental Papers 2022(016):A001. https://doi.org/10.5089/9798400218507.087.A001
- Bateki CA, Wassie SE, Wilkes A (2023) The contribution of livestock to climate change mitigation: a perspective from a low-income country. Carbon Management 14(1):1–16. https://doi.org/10.1080/ 17583004.2023.2173655
- Bedasa Y, Deksisa K (2024) Food insecurity in East Africa: an integrated strategy to address climate change impact and violence conflict. J Agriculture Food Res 15:100978. https://doi.org/10.1016/j. jafr.2024.100978
- Bede-Ojimadu O, Orisakwe OE (2020) Exposure to wood smoke and associated health effects in Sub-Saharan Africa: a systematic review. Ann Glob Health 86(1):32. https://doi.org/10.5334/aogh. 2725
- Behera B, Ali A (2016) Patterns and determinants of household use of fuels for cooking: empirical evidence from sub-Saharan Africa. Energy 117:93–104. https://doi.org/10.1016/j.energy.2016.10.055
- Berhe TG, Tesfahuney RG, Desta GA, Mekonnen LS (2017) Biogas plant distribution for rural household sustainable energy supply in Africa. Energy and Policy Research 4(1):10–20. https://doi.org/ 10.1080/23815639.2017.1280432
- Bikorimana D, Mupenzi C, Mudatsikira V (2023) Deforestation driven by agriculture cash crops, animal husbandry, and population growth in Rwanda (1992–2018). International Journal of Advances in Engineering and Management 5(1):101–114. https://doi.org/10.35629/5252-0501101114.
 Retrieved January 27 2024 from https://ijaem.net/issue_dcp/Deforestation%20Driven%20by% 20Agriculture%20Cash%20Crops,%20Animal%20Husbandry,%20and%20Population%20Gro wth%20in%20Rwanda%20(1992%20to%202018)..pdf
- Brini R (2021) Renewable and non-renewable electricity consumption, economic growth and climate change: evidence from a panel of selected African countries. Energy, 223. https://doi.org/10. 1016/j.energy.2021.120064
- Bucagu C, Vanlauwe B, Van Wijk MT, Giller KE (2014) Resource use and food self-sufficiency at farm scale within two agro-ecological zones of Rwanda. Food Security 6(5):609–628. https://doi.org/ 10.1007/s12571-014-0382-0

- Chen D, Mechlowitz K, Li X, Schaefer N, Havelaar AH, McKune SL (2021) Benefits and risks of smallholder livestock production on child nutrition in low- and middle-income countries [Review]. Frontiers in Nutrition, 8. https://doi.org/10.3389/fnut.2021.751686
- Christiaensen L, Demery L, Kuhl J (2011) The (evolving) role of agriculture in poverty reduction an empirical perspective. J Dev Econ 96:239–254. https://doi.org/10.1016/j.jdeveco.2010.10.006
- Clemens H, Bailis R, Nyambane A, Ndung'u V (2018) Africa Biogas Partnership Program: a review of clean cooking implementation through market development in East Africa. Energy Sustain Dev 46:23–31. https://doi.org/10.1016/j.esd.2018.05.012
- Collishaw A, Janzen S, Mullally C, Camilli H (2023) A review of livestock development interventions' impacts on household welfare in low- and middle-income countries. Glob Food Sec 38:100704. https://doi.org/10.1016/j.gfs.2023.100704
- Corral S, Díaz A, Monagas M, García E (2017) Agricultural policies and their impact on poverty reduction in developing countries: lessons learned from three water basins in Cape Verde. Sustainability 9(1841). https://doi.org/10.3390/su9101841
- Dagnachew AG, Hof AF, Lucas PL, van Vuuren DP (2020) Scenario analysis for promoting clean cooking in Sub-Saharan Africa: costs and benefits. Energy 192. https://doi.org/10.1016/j.energy.2019.116641
- Erdaw MM (2023) Contribution, prospects and trends of livestock production in sub-Saharan Africa: a review. Int J Agric Sustain 21(1):2247776. https://doi.org/10.1080/14735903.2023.2247776
- Ezeanya C, Kennedy A (2016) Integrating clean energy use in national poverty reduction strategies: opportunities and challenges in Rwanda's Girinka programme. WIDER Working Paper 2016/23. In. Helsinki: UNU-WIDER. https://doi.org/10.35188/UNU-WIDER/2016/066-9
- Ezeanya C, Kennedy A (2017) Integrating clean energy use in national poverty reduction strategies: opportunities and challenges in Rwanda's Girinka Programme. In The Political Economy of Clean Energy Transitions (pp. 0). Oxford University Press. https://doi.org/10.1093/oso/9780198802242.003.0023
- FAO (2018) Ending poverty and hunger by investing in agriculture and rural areas. Rome: Food and Agriculture Organization of the United Nations. Retrieved 29 May 2018 from https://www.fao.org/3/ 17556EN/i7556en.pdf
- Garfi M, Marti-Herrero J, Garwood A, Ferrer I (2016) Household anaerobic digesters for biogas production in Lain America: a review. Renew Sustain Energy Rev 60:599–614. https://doi.org/10.1016/j.rser. 2016.01.071
- Gashure S, Wana D, Samimi C (2022) Impacts of climate variability and climate-smart agricultural practices on crop production in UNESCO designated cultural landscapes of Konso. Ethiopia Theoretical App Climatol 150(3–4):1495–1511. https://doi.org/10.1007/s00704-022-04244-9
- Geddafa T, Melka Y, Sime G (2023) Cost-benefit analysis and financial viability of household biogas plant investment in South Ethiopia. Sustainable Energy Res 10(1):20. https://doi.org/10.1186/ s40807-023-00089-6
- Gitau K, Mutune J, Sundberg C, Mendum R, Njenga M (2019) Factors influencing the adoption of biocharproducing gasifier cookstoves by households in rural Kenya. Energy Sustain Dev 52:63–71. https://doi. org/10.1016/j.esd.2019.07.006
- Gonzáles C, Schiek BE, Mwendia SW, Prager SD (2016) Improved forages and milk production in East Africa. A case study in the series: economic foresight for understanding the role of investments in agriculture for the global food system. Centro Internacional de Agricultura Tropical. Publication No. 422. Retrieved 29 May 2018 from https://hdl.handle.net/10568/77557
- Guo H, Xia Y, Jin J, Pan C (2022) The impact of climate change on the efficiency of agricultural production in the world's main agricultural regions. Environ Impact Assess Rev 97:106891. https://doi.org/10. 1016/j.eiar.2022.106891
- Habiyaremye N, Ouma EA, Mtimet N, Obare GA (2021) A review of the evolution of dairy policies and regulations in Rwanda and its implications on inputs and services delivery. Frontiers Vet Sci. 8, Article 611298. https://doi.org/10.3389/fvets.2021.611298
- Hansen J, Hellin J, Rosenstock T, Fisher E, Cairns J, Stirling C, Lamanna C, van Etten J, Rose A, Campbell B (2019) Climate risk management and rural poverty reduction. Agric Syst 172:28–46. https://doi. org/10.1016/j.agsy.2018.01.019
- IFAD (2016) Rwanda dairy development project: detailed design report. Retrieved 27 May 2018 from https://webapps.ifad.org/members/eb/118/docs/EB-2016-118-R-19-Project-design-report.pdf
- IFC (2007) Lease a cow? IFC helps make it possible. International Finance Corporation, World Bank Group. Retrieved 23 May 2018 from www.ifc.org/wps/wcm/connect/news_ext_content/ifc_external_ corporate_site/news+and+events/news/rwand_cow_leasing
- IFRC (2016) Cost-benefit analysis case study: livestock rotation programme in Rwanda. Retrieved 23 May 2018 from https://www.livelihoodscentre.org/-/cost-benefit-analysis-cba-case-study-rwanda-lives tock-rotation-programme

- Inoni O (2010) Effects of small-holder livestock production in reducing rural poverty in the central agricultural zone of Delta State, Nigeria. Libyan Agriculture Res Center J Int 1(3):171–176
- IPCC (2007) Fourth assessment report: synthesis. Retrieved 27 May 2018 from www.ipcc.ch/ipccreports/ ar4-syr.htm
- IPCC (2023) Climate change 2023: synthesis report. Contribution of working groups I, II and III to the sixth assessment report of the intergovernmental panel on climate change. H. Lee and J. Romero (eds.). IPCC, Geneva, Switzerland, pp. 1–34. Retrieved 27 May 2018 from https://doi.org/10.59327/IPCC/ AR6-9789291691647.001
- Iraguha B, Hamudikuwanda H, Mushonga B (2015) Bovine mastitis prevalence and associated risk factors in dairy cows in Nyagatare District, Rwanda. J S Afr Vet Assoc 86(1):1–6
- Issahaku M, Derkyi NSA, Kemausuor F (2024) A systematic review of the design considerations for the operation and maintenance of small-scale biogas digesters. Heliyon. https://doi.org/10.1016/j.heliyon. 2024.e24019
- Jagger P, Das I (2018) Implementation and scale-up of a biomass pellet and improved cookstove enterprise in Rwanda. Energy Sustain Dev 46:32–41
- Juozaitiene V, Juozaitis A, Micikeviciene R (2006) Relationship between somatic cell count and milk production or morphological traits of udder in Black-and-White cows. Turkish J Vet Animal Sci. 30(1):47–51. Retrieved 15 June 2018 from https://journals.tubitak.gov.tr/veterinary/vol30/iss1/7
- Kafle K (2014) Is there more than milk? The impact of Heifer International's livestock donation program on rural livelihoods: preliminary findings from a field experiment in Zambia. Agricultural Applied Econ Ass Minneapolis, Minnesota. Retrieved 29 May 2018 from https://ageconsearch.umn.edu/ record/170629/files/Livestock-Zambia-KashiR.pdf
- Kalina M, Ogwang JÒ, Tilley E (2022) From potential to practice: rethinking Africa's biogas revolution. Humanities Soc Sci Comm 9(1):374. https://doi.org/10.1057/s41599-022-01396-x
- Kandulu JM, Connor JD (2017) Improving the effectiveness of aid: an evaluation of prospective Mekong irrigation investments. Int J Water Resour Dev 33(2):270–291. https://doi.org/10.1080/07900627. 2016.1188060
- Kayigema V, Rugege D (2014) Women's perceptions of the Girinka (one cow per poor family) programme, poverty alleviation and climate resilience in Rwanda. Agenda 28(3):53–64. https://doi. org/10.1080/10130950.2014.939839
- Kayigema V (2013) An assessment of the Girinka (one cow per poor family) program and poverty alleviation in Rwanda a case study of Bugesera District University of KwaZulu-Natal]. Doctoral dissertation. http://hdl.handle.net/10413/11829. Retrieved 29 May 2018 from https://researchspace. ukzn.ac.za/bitstreams/d96dde3f-9d94-4d29-8d9f-f28841e8c107/download
- Keerthana Devi M, Manikandan S, Oviyapriya M, Selvaraj M, Assiri MA, Vickram S, Subbaiya R, Karmegam N, Ravindran B, Chang SW, Awasthi MK (2022) Recent advances in biogas production using Agro-Industrial Waste: a comprehensive review outlook of Techno-Economic analysis. Biores Technol 363:127871. https://doi.org/10.1016/j.biortech.2022.127871
- Khundi-Mkomba F, Saha AK, Wali UG (2023) Social protection schemes in rural Rwanda: a panacea for household energy burdens? Energy Sustain Dev 74:91–101. https://doi.org/10.1016/j.esd.2023.03.006
- Kim SK, Tiessen K, Beeche A, Mukankurunziza J, Kamatari A (2011) Cattle manure management in Rwanda – a case of Girinka cow beneficiaries in the district of Ngoma. Agric Sci 24:39–48 Retrieved 29 May 2018 from https://www.ajol.info/index.php/rj/article/download/76308/66766/0
- Kim S, Tiessen K, Beeche A, Kamatari A (2013) Soil fertility and manure management: lessons from the knowledge, attitudes and practices of Girinka farmers in the district of Ngoma. Rwanda Agroecol Sustain Food Systems 37(6):631–658. https://doi.org/10.1080/21683565.2012.762636
- Klapwijk CJ, Bucagu C, van Wijk MT, Udo HMJ, Vanlauwe B, Munyanziza E, Giller KE (2014) The 'One cow per poor family' programme: current and potential fodder availability within smallholder farming systems in southwest Rwanda. Agric Syst 131:11–22. https://doi.org/10.1016/j.agsy.2014. 07.005
- Li J, Ma W, Zhu H (2023) A systematic literature review of factors influencing the adoption of climate-smart agricultural practices. Mitig Adapt Strat Glob Change 29(1):2. https://doi.org/10.1007/s11027-023-10098-x
- Lwiza F, Mugisha J, Walekhwa PN, Smith J, Balana B (2017) Dis-adoption of household biogas technologies in Central Uganda. Energy Sustain Dev 37:124–132. https://doi.org/10.1016/j.esd.2017.01.006
- Madamombe SM, Ng'ang'a SK, Öborn I, Nyamadzawo G, Chirinda N, Kihara J, Nkurunziza L (2024) Climate change awareness and adaptation strategies by smallholder farmers in semi-arid areas of Zimbabwe. Int J Agric Sustain 22(1):2293588. https://doi.org/10.1080/14735903.2023.2293588
- Manero A, Bjornlund H, Wheeler S, Zuo A, Mdemu M, Van Rooyen A, Chilundo M (2020) Growth and inequality at the micro scale: an empirical analysis of farm incomes within smallholder irrigation

systems in Zimbabwe, Tanzania and Mozambique. Int J Water Resour Dev 36(Suppl. 1):S224–S245. https://doi.org/10.1080/07900627.2020.1811959

- Manzi M, Rydhmer L, Ntawubizi M, D'Andre Hirwa C, Karege C, Strandberg E (2020) Milk production and lactation length in Ankole cattle and Ankole crossbreds in Rwanda. Trop Anim Health Prod 52(6):2937–2943. https://doi.org/10.1007/s11250-020-02311-9
- Miklyaev M, Afra S, Schultz M (2017) Cost-benefit analysis of Rwanda's dairy value chains. Development Discussion Papers, 2. Retrieved 29 May 2018 from https://cri-world.com/publications/qed_ dp_299.pdf
- MINAGRI (2006) A cow to each poor family in Rwanda. Ministry of Agriculture proposal, Kigali, Rwanda. Retrieved 29 May 2018 from https://www.minagri.gov.rw/index.php?eID=dumpFile&t= f&f=16983
- Mperejekumana P, Shen L, Zhong S, Muhirwa F, Gaballah MS, Nsigayehe JMV (2024) Integrating climate change adaptation into water-energy-food-environment nexus for sustainable development in East African Community. J Clean Prod 434:140026. https://doi.org/10.1016/j.jclepro.2023.140026
- Mubenga-Tshitaka J-L, Dikgang J, Muteba Mwamba JW, Gelo D (2023) Climate variability impacts on agricultural output in East Africa. Cogent Econ Finance 11(1):2181281. https://doi.org/10.1080/ 23322039.2023.2181281
- Mukeshimana MC, Zhao Z-Y, Ahmad M, Irfan M (2021) Analysis on barriers to biogas dissemination in Rwanda: AHP approach. Renewable Energy 163:1127–1137. https://doi.org/10.1016/j.renene. 2020.09.051
- Mutenje MJ, Farnworth CR, Stirling C, Thierfelder C, Mupangwa W, Nyagumbo I (2019) A cost-benefit analysis of climate-smart agriculture options in Southern Africa: balancing gender and technology. Ecol Econ 163:126–137. https://doi.org/10.1016/j.ecolecon.2019.05.013
- Mutimura M, Everson T (2011) Assessment of livestock feed resource-use patterns in low rainfall and aluminium toxicity prone areas of Rwanda. African J Agricutural Res 6(15):3461–3469. https://doi.org/ 10.5897/AJAR10.315
- Mwabonimana MF, Rekeraho J, Ntirenganya M, Bimenyimana F (2015) Cost benefit analysis of bovine mastitis in Nyabihu and Musanze districts, Rwanda. Int J Novel Res Life Sci 2(4):21–31. Available at https://www.noveltyjournals.com/upload/paper/Cost%20Benefit%20Analysis%20of%20Bovine-293. pdf
- Mwakaje AG (2008) Dairy farming and biogas use in Rungwe district, South-west Tanzania: a study of opportunities and constraints. Renew Sust Energ Rev 12(8):2240–2252. https://doi.org/10.1016/j.rser. 2007.04.013
- Nilsson P (2019) The role of land use consolidation in improving crop yields among farm households in Rwanda. J Develop Stud 55(8):1726–1740. https://doi.org/10.1080/00220388.2018.1520217
- Nilsson P, Backman M, Bjerke L, Maniriho A (2019) One cow per poor family: effects on the growth of consumption and crop production. World Dev 114:1–12. https://doi.org/10.1016/j.worlddev.2018.09. 024
- Nix E, Fleeman N, Lorenzetti F, Lewis J, Shen G, Abebe L, Li Y, Xing R, Maden M, Duarte R, Bresnahan R, Williams KN, Adair-Rohani H, Pope D, Puzzolo E (2022) Health effects of liquid and gaseous fuels for household energy use: systematic evidence mapping. Environ Res Lett 17(12):123003. https://doi.org/10.1088/1748-9326/aca1d2
- Niyitanga A, Kabayiza A, Nitonzima J (2015) Assessment of yield gaps in main staple crops in Rwanda. Int J Agriculture Innov Res 3(4):1267–1271
- Njenga M, Sears RR, Mendum R (2023) Sustainable woodfuel systems: a theory of change for sub-Saharan Africa. Environ Res Comm 5(5):051003. https://doi.org/10.1088/2515-7620/acd0f3
- Njoya HM, Matavel CE, Msangi HA, Wouapi HAN, Löhr K, Sieber S (2022) Climate change vulnerability and smallholder farmers' adaptive responses in the semi-arid Far North Region of Cameroon. Discover Sustainability 3(1):41. https://doi.org/10.1007/s43621-022-00106-6
- Ogisi OD, Begho T (2023) Adoption of climate-smart agricultural practices in sub-Saharan Africa: a review of the progress, barriers, gender differences and recommendations. Farming System 1(2):100019. https://doi.org/10.1016/j.farsys.2023.100019
- Ojara MA, Lou YS, Babaousmail H, Sempa AK, Ayugi B, Ogwang BA (2022) Evaluation of drought, wet events, and climate variability impacts on maize crop yields in East Africa during 1981–2017. Int J Plant Production 16(1):41–62. https://doi.org/10.1007/s42106-021-00178-w
- Omotoso AB, Letsoalo S, Olagunju KO, Tshwene CS, Omotayo AO (2023) Climate change and variability in sub-Saharan Africa: a systematic review of trends and impacts on agriculture. J Clean Prod 414:137487. https://doi.org/10.1016/j.jclepro.2023.137487
- Onyekaozuoro EC, Daim TU, Herstatt C (2023). Roadmapping of biogas production technology in Sub-Saharan Africa: waste to energy. In T. U. Daim, R. Phaal, D. Meissner, & C. Kerr (Eds.), Next

Generation Roadmapping: Establishing Technology and Innovation Pathways Towards Sustainable Value (pp. 181–222). Springer, Cham. London, UK. https://doi.org/10.1007/978-3-031-38575-9_8

- Ortiz-Bobea A, Ault TR, Carrillo CM, Chambers RG, Lobel DB (2021) Anthropogenic climate change has slowed global agricultural productivity growth. Nat Clim Chang 11(4):306-U328. https://doi.org/10. 1038/s41558-021-01000-1
- Overfield D (1998) An investigation of the household economy: coffee production and gender relations in Papua New Guinea. J Develop Stud 34(5):52. https://doi.org/10.1080/00220389808422536
- Paul BK, Frelat R, Birnholz C, Ebong C, Gahigi A, Groot JCJ, Herrero M, Kagabo DM, Notenbaert A, Vanlauwe B, van Wijk MT (2017) Agricultural intensification scenarios, household food availability and greenhouse gas emissions in Rwanda: Ex-ante impacts and trade-offs. Agric Syst. https://doi.org/ 10.1016/j.agsy.2017.02.007
- Rafael BM (2023) The importance of agricultural development projects: a focus on sustenance and employment creation in Kenya, Malawi, Namibia, Rwanda, and Uganda. J Agric Chem Environ 12(2):152–170
- Rasimphi TE, Tinarwo D (2020) Relevance of biogas technology to Vhembe district of the Limpopo province in South Africa. Biotechnology Reports 25:e00412. https://doi.org/10.1016/j.btre.2019.e00412
- Rawlins R, Pimkina S, Barrett C, Pedersen S, Wydick B (2014) Got milk? The impact of Heifer International's livestock donation programs in Rwanda on nutritional outcomes. Food Policy 44:202–213. https://doi. org/10.1016/j.foodpol.2013.12.003
- RGB (2018) Assessing Girinka progamme (2006–2016): citizen perspectives (Citizen Report Card, Issue. Rwanda Governance Board. Retrieved 27 February 2019 from http://rgb.rw/fileadmin/Key_docum ents/HGS/GIRINKA_REPORT_2018.pdf
- Robinson LA, Hammitt JK (2017) Benefit-cost analysis in global health. Forthcoming as Chapter 7 in: Global Health Priority-Setting: Beyond Cost-Effectiveness. In: Norheim O et al., (eds) Oxford, UK and New York, NY: Oxford University Press. Retrieved 27 May 2018 from https://srn.com/abstract=2952014
- Rocha-Meneses L, Luna-delRisco M, González CA, Moncada SV, Moreno A, Sierra-Del Rio J, Castillo-Meza LE (2023) An overview of the socio-economic, technological, and environmental opportunities and challenges for renewable energy generation from residual biomass: a case study of biogas production in Colombia. Energies 16(16):5901. Retrieved 27 January 2024 from https://www.mdpi.com/1996-1073/ 16/16/5901
- Roopnarain A, Adeleke R (2017) Current status, hurdles and future prospects of biogas digestion technology in Africa. Renew Sustain Energy Rev 67:1162–1179. https://doi.org/10.1016/j.rser.2016.09.087
- Rubagumya I, Komakech AJ, Kabenge I, Kiggundu N (2023) Potential of organic waste to energy and biofertilizer production in Sub-Saharan Africa: a review. Waste Disposal Sustain Energy 5(3):259–267. https://doi.org/10.1007/s42768-022-00131-1
- Salazar L, Fahsbender J, Kim N (2018) Livestock transfers, food security and women's empowerment: evidence from a randomized phased-in program in Nicaragua. Washington, DC. https://doi.org/10. 18235/0001447. Retrieved 27 March 2019 from https://publications.iadb.org/publications/english/ viewer/Livestock_Transfers_Food_Security_Womens_Empowerment_Nicaragua_final.pdf
- Sapp AC, Nane GF, Amaya MP, Niyonzima E, Hategekimana JP, VanSickle JJ, Gordon RM, Havelaar AH (2023) Estimates of disease burden caused by foodborne pathogens in contaminated dairy products in Rwanda. BMC Public Health 23(1):657. https://doi.org/10.1186/s12889-023-15204-x
- Sepee S, Tesfahun AA (2023) Energy, gender and development: the impact of energy efficient cookstoves intervention on the welfare of women in Ethiopia. Int J Energy Sect Manage. https://doi.org/10.1108/ IJESM-01-2023-0021
- Shackleton C, Sinasson G, Adeyemi O, Martins V (2022) Fuelwood in South Africa revisited: widespread use in a policy vacuum. Sustainability 14(17):11018. Retrieved 27 January 2023 from https://www. mdpi.com/2071-1050/14/17/11018
- Shikuku KM, Valdivia RO, Paul BK, Mwongera C, Winowiecki L, Läderach P, Herrero M, Silvestri S (2017) Prioritizing climate-smart livestock technologies in rural Tanzania: a minimum data approach. Agric Syst 151:204–216. https://doi.org/10.1016/j.agsy.2016.06.004
- Surendra KC, Takara D, Hashimoto AG, Khanal SK (2014) Biogas as a sustainable energy source for developing countries: opportunities and challenges. Renew Sust Energ Rev 31:846–859. https://doi. org/10.1016/j.rser.2013.12.015
- Swarnam TP, Velmurugan A, Subramani T, Ravisankar N, Subash N, Pawar AS, Perumal P, Jaisankar I, Dam Roy S (2024) Climate smart crop-livestock integrated farming as a sustainable agricultural strategy for humid tropical islands. Int J Agric Sustain 22(1):2298189. https://doi.org/10.1080/14735 903.2023.2298189
- Tabe-Ojong MP Jr, Kedinga ME, Gebrekidan BH (2024) Behavioural factors matter for the adoption of climate-smart agriculture. Sci Rep 14(1):798. https://doi.org/10.1038/s41598-023-50264-4

- UN General Assembly (2015) UN General Assembly. Resolution adopted by the General Assembly on 25 September 2015. Transforming our world: the 2030 Agenda for Sustainable Development. Retrieved 27 May 2018 from https://www.un.org/en/development/desa/population/migration/generalassembly/ docs/globalcompact/A_RES_70_1_E.pdf
- USAID Rwanda (2016) Feed the future innovation lab for livestock systems Rwanda: Animal source foods production and marketing brief. Livestock Lab. Retrieved 31 May 2018 from https://livestocklab.ifas.ufl.edu/media/livestocklabifasufledu/pdf-/pdfs-by-country-pre2019/RwandaBriefASFP rodMktfinal.pdf
- Vásquez P, Gallego V, Soto JD (2024) Transforming MSMEs towards circularity: an attainable challenge with the appropriate technologies and approaches. Environment Systems Decisions. https://doi.org/ 10.1007/s10669-023-09961-8
- Warinda E, Nyariki DM, Wambua S, Muasya RM, Hanjra MA (2020) Sustainable development in East Africa: impact evaluation of regional agricultural development projects in Burundi, Kenya, Rwanda, Tanzania, and Uganda. Nat Res Forum 44(1):3–39. https://doi.org/10.1111/1477-8947.12191
- Wilkes A, Wassie S, Odhong C, Fraval S, van Dijk S (2020) Variation in the carbon footprint of milk production on smallholder dairy farms in central Kenya. J Clean Prod 265:121780. https://doi.org/10. 1016/j.jclepro.2020.121780
- World Bank (2016) Poverty and shared prosperity 2016: taking on inequality. The World Bank. Retrieved 15 June 2018. https://doi.org/10.1596/978-1-4648-0958-3
- World Bank (2018) Poverty and shared prosperity 2018: piecing together the poverty puzzle. Retrieved 15 June 2018. https://doi.org/10.1596/978-1-4648-1330-6
- Wright MM, Auldist MJ, Kennedy E, Dunshea FR, Hannah M, Wales WJ (2016) Variation in feeding behavior and milk production among dairy cows when supplemented with 2 amounts of mixed ration in combination with 2 amounts of pasture. J Dairy Sci 99(8):6507–6518. https://doi.org/10.3168/jds. 2015-10771

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.